Author's response to reviews

Title: Bronchogenic cyst of ileal mesentery: a case report and review of literature

Authors:

Adolfo Petrina (adolfo.petrina@poste.it)
Carlo Boselli (carloboselli@yahoo.it)
Roberto Cirocchi (cirocchiroberto@yahoo.it)
Piero Covarelli (piero.covarelli@med.unipg.it)
Emilio Eugeni (emilioeugeni@libero.it)
Marco Badolato (drdionysos@hotmail.com)
Luigi Finocchi (luigifinocchi@yahoo.it)
Giuseppe Noya (gnoya@unipg.it)

Version: 2 Date: 13 July 2010

Author's response to reviews:

Abstract now structured in three sections: Introduction, Case presentation and Conclusions; removed dates which might identify the patient; highlighted changes made in the revised manuscript; waiting for your decision!

Best regards,
adolfo petrina